Author's response to reviews

Title: Giant hepatic hydatid cyst with subfascial extension treated by open minimal invasive surgery: a case report

Authors:

Dipesh Duttaroy (drduttaroy@gmail.com)
Samir Kacheriwala (dr_samir_k@yahoo.com)
Bithika Duttaroy (drbithika@gmail.com)
Jitendra Jagtap (drijitendrajagtap@gmail.com)
Gunjan Patel (gunjanap@yahoo.com)
Nikhil Modi (drnjmodi@yahoo.com)

Version: 2 Date: 6 September 2007

Author's response to reviews:

Dear Sir,

With reference to your email I am resubmitting the article with the suggested changes.
- the title has been restructured so that it ends in ': a case report.'
- ‘Title page’ subheading has been removed
- All details from the title page other than the title, author details and corresponding author address have been removed.
- the occupation of the patient has been deleted from the manuscript
- the keywords section has been removed
- only 3 figures are being submitted.

Thanking You,

Dr D Duttaroy